Wed Cct 21 16:49: 07 GVI 1998
Year: 98 Doy: 294
Cbserver: el nore

Wed Cct 21 16:50:55 GVII 1998 CH P Startup--Initializing new tape
Wed Cct 21 16:55:22 GVIT 1998 CH P CH P Start Patrol

Wed Oct 21 16:58:34 GV 1998 CH P CH P End Patr ol

Wed Cct 21 16:59:24 GMTI 1998 CH P CH P Start Patrol

Wed Cct 21 17:02:57 GMI 1998 CH P Bi as

Wed Cct 21 17:03:48 GMTI 1998 CH P End Bi as

Wed Oct 21 17:03:56 GVII 1998 CH P Wat er

Wed COct 21 17:04:32 GVIT 1998 CH P End Water

COMMENT: Wed COct 21 17:15:48 GMI 1998

Darryl read the HELCO neters this norning.

Wed Oct 21 17:22:33 GV 1998 CH P CH P End Patrol
Wed Oct 21 17:23:21 GVII 1998 CH P CH P Start Patrol
WEATHER COMMVENT: Wed Cct 21 17:42:15 GMr 1998

Crrus covering all the sky with varying thickness

CH P COMMENT: Wed Oct 21 17:42:53 GMIr 1998

CH P |ives!

POLARCOR pol ari zers have replaced Sanritz pol ari zers

in the Lyot filter. Al 5 polarizers replaced.

Exposure tinme had to be reduced from 30nsec

to 10nsec, even through cirrus. It is expected
that it will be shortened even nore when full flux
is seen.

Al so repaired was the filter tenperature sensor.
Tenperature control is obviously working.

Wed Cct 21 18:02:49 GMI 1998 CH P Bi as
Wed Cct 21 18:03:47 GMI 1998 CH P End Bi as
Wed Cct 21 18:03:54 GMI 1998 CH P Wat er
Wed Cct 21 18:04: 33 GVII 1998 CH P End Water

WEATHER COWMMENT: Wed Cct 21 18:57:13 GMTI 1998
Cirrus getting really thick.

Wed Cct 21 19:02: 00 GVI 1998 CH P Bi as

Wed Cct 21 19:02:58 GMI 1998 CH P End Bi as

Wed Cct 21 19:03: 06 GMI 1998 CH P Wat er

Wed Cct 21 19:03:42 GVII 1998 CH P End Water

Wed Cct 21 19:12:28 GVII 1998 CH P CH P End Patr ol
Wed Cct 21 19:12:58 GMI 1998 CH P endi ng tape

Wed Cct 21 19:24:17 GMI 1998 WKI
Wed Cct 21 19:35:53 GMI 1998
Wed Cct 21 19:36:29 GMI 1998
Wed Cct 21 19:48:24 GVII 1998
Wed Cct 21 20:12:34 GVIT 1998
Wed Cct 21 21:29:44 GMI 1998

Il Start Patrol
Il Start Patrol
Il End Patr ol

Il Start Patrol
Il Start Patrol
Il Start Patrol

X5%5%



Wed Cct 21 21:30: 08 GMI 1998 MKI T End Patro

MKI Il COMMENT: Wed Oct 21 22:57:54 GMIr 1998

Work continues on mounting and testing the new MKIV camera and equi pnent.
COWMMENT: Wed Oct 21 22:59:11 GMTI 1998

TAPES:
CH P: C00708
LOW.: LOO0549 in drive #1

Wed Cct 21 23:19:30 GVII 1998
M1 T

19 37.rawrk3 20_12.rawnk3 20 _24.rawrk3 20_37.rawnk3 21_20.rawrk3
19 48.rawrk3 20_15.rawnk3 20_27.rawrk3 20_40.rawnk3 21_23.rawrk3
19 53. rawmk3 20 18.rawnk3 20 31.rawrk3 20_43.rawrk3
19 56.rawnk3 20 21.rawnk3 20 34.rawrk3 21 17.rawrk3



